MATHEMATICAL TRIPOS Part III

Wednesday, 2 June, 2010 9:00 am to 11:00 am

PAPER 69

POLAR OCEANS AND CLIMATE CHANGE

Attempt no more than THREE questions.

There are FOUR questions in total.

The questions carry equal weight.

In all questions assume that the latent heat of fusion of ice is 336 kJ kg^{−1}, and its specific heat is 2.1 kJ kg^{−1}K^{−1}.

The psu, or practical salinity unit, is equivalent to 1 part per thousand.

STATIONERY REQUIREMENTS SPECIAL REQUIREMENTS Cover sheet None Treasury Tag Script paper

You may not start to read the questions printed on the subsequent pages until instructed to do so by the Invigilator.

Part III, Paper 69

Part III, Paper 69 [TURN OVER

END OF PAPER

Part III, Paper 69